

Delft University of Technology

Does the availability of alternative energy choices lead to more environmentally friendly outcomes? The case of thermal energy communities and natural gas consumption

Fouladvand, Javanshir; Ates, Emre; Sarı, Yasin; Okur, Özge

DOI 10.1016/j.apenergy.2024.123932

Publication date 2024 **Document Version** Final published version

Published in Applied Energy

Citation (APA)

Fouladvand, J., Ateş, E., Sarı, Y., & Okur, Ö. (2024). Does the availability of alternative energy choices lead to more environmentally friendly outcomes? The case of thermal energy communities and natural gas consumption. *Applied Energy*, *374*, Article 123932. https://doi.org/10.1016/j.apenergy.2024.123932

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.



Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

Does the availability of alternative energy choices lead to more environmentally friendly outcomes? The case of thermal energy communities and natural gas consumption

Javanshir Fouladvand^{a,*}, Emre Ateş^b, Yasin Sarı^c, Özge Okur^b

^a Copernicus Institute of Sustainable Development, Utrecht University

^b Multi-Actor Systems department, Technology, Policy and Management faculty, Delft University of Technology (TU Delft)

^c Enterprise Application Systems, Group IT, Nationale-Nederlanden (NN Group)

HIGHLIGHTS

• Agent-based modelling and simulation is used to study thermal energy transition.

• Multi-level governance is used as a theoretical backbone in an agent-based model.

• The importance of the availability of alternative energy choices is demonstrated.

• The availability of several natural gas sources leads to more sustainable outcomes.

• The establishment of thermal energy communities is vital for the energy transition.

ARTICLE INFO

ABSTRACT

Keywords: Energy transition Thermal energy community Agent-based modelling and simulation Collective action Multi-level governance Natural gas Individual households, responsible for 25% of total energy consumption in Europe, are crucial actors in the energy transition. Although various policies and energy choices are available for such actors (e.g., individual solar photovoltaic or community energy systems), they are usually restricted to municipal governance, and public opinion towards national energy policy is not strongly presented. This study explores and describes the influence of the availability of alternative energy choices on different levels of governance to facilitate households' energy transition. An agent-based model is conceptualised through multi-level governance, the institutional analysis and development framework and the social value theory. To also address the ongoing energy crisis, the study focuses on the Dutch thermal energy transition and the thermal energy communities as a collective action for generating, distributing, and consuming renewable thermal energy and, therefore, three layers of energy choice alternatives are provided for households: national level (i.e., sources of natural gas), municipal/ community level (i.e., collective renewable energy technologies), and individual level. The results delineated the importance of the availability of alternative energy choices in the suggested multi-level governance collective action system. Such systems consumed only 12% natural gas, while they covered their thermal demand by increasing the capacity of collective thermal energy systems (83% on average) and adopting more individual thermal energy systems (heat pump, approximately 85%). Although the performances on voluntary blackouts/ discomfort (2.7% on average) and CO₂ emissions (85% reduction approximately) were also positive, this was reflected in a significant price increase.

1. Introduction

Addressing climate change and the pivotal need to reduce greenhouse gas emissions (GHG, such as CO_2) requires significant transformation in the energy sector [1]. Such transformation, mainly referred

to as the energy transition, has various facades, such as energy demand reduction strategies, energy efficiency and deployment of renewable energy technologies (RETs), which include and influence different actors (e.g., individual households, industries and businesses) across the globe [2].

* Corresponding author. E-mail addresses: j.fouladvand@uu.nl, javanshir.fouladvand.work@gmail.com (J. Fouladvand).

https://doi.org/10.1016/j.apenergy.2024.123932

Received 17 February 2024; Received in revised form 1 June 2024; Accepted 12 July 2024 Available online 2 August 2024

0306-2619/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Individual households are responsible for approximately one-fourth of total energy consumption in Europe [3], making their participation crucial for a successful energy transition [4]. However, as illustrated in studies such as [5,6], it is impossible to suggest a one-size-fits-all solution that facilitates the households' energy transition due to different reasons, such as heterogeneity in their motives and demands. Therefore, various studies explored the behaviour and preferences of individual households regarding different energy sources and strategies.

From the individual households' perspective, the influence and importance of values on evaluating different energy alternatives and adopting them in the Netherlands are investigated in [7]. By employing behavioural economics and exploring the influence of psychological and behavioural conditions, the households' energy-related decision-making and behaviour are studied in [8]. The importance of the attitude and knowledge of saving behaviours in the residential sector is explored in [9]. Dutch households' energy efficiency decisions and CO₂ emission reduction strategies are explored in [10], showing the importance of natural gas prices in adopting energy-efficient technologies. Factors influencing energy efficiency investments in existing Swedish residential buildings are explored in [11], and the influence of knowledge and economic incentives for such decisions is highlighted. The adoption of renewable heating systems in Italy is investigated in [12] by employing the diffusion of innovation theory. Studies such as [12-14] explored the influence of user profiles, socio-economic factors and values on households' energy consumption. In addition, various studies have focused on the influence of policies and regulations on households' energy transition. The influence of decarbonisation policies in the residential sector in central and southern eastern Europe is studied in [15]. The impact of taxing households' energy consumption in Spain is investigated in [16]. Along with such studies focusing on individual households' behaviours, various studies studied households' collective action for energy generation and consumption within their neighbourhood, namely community energy systems (CESs).

As an overarching term, CES is defined based on the collective action of individual households to generate, distribute and consume renewable energy resources (RECs) [17,18]. In this context, by collecting empirical data, studies such as [19,20] explore the role of institutional and sociopsychological factors in participating in CESs. The perceptions towards natural gas consumption and import in Europe are presented in [21], and it highlights the importance of knowing the natural gas sources for households. Factors influencing the adoption of renewable heating systems and participating in (thermal) energy communities are studied in [22]. The role of behaviour and leadership in the emergence of CESs in the Dutch context is explored in [23]. Studies such as [24,25] investigated the role of renewable heating systems, namely thermal energy communities (TEC initiatives), in the heat transition in Germany and the Netherlands. The influence of the different institutional factors (e.g., regulations and policies) and internal and external networks (e.g., relationships and dynamics within the neighbourhoods) are investigated in [20,26]. By employing computational social simulation approaches such as agent-based modelling and simulation (ABMS), [27,28] explore different settings for the collective decision-making processes for TEC initiatives, demonstrating the potential of such systems in reducing natural gas consumption. The influence of different technical, institutional and behavioural settings on the energy security of CESs and TEC initiatives are also explored in [29,30]. Therefore, this branch of literature, namely the (individual and collective) behaviour of households, is expanding, and the participation of households in the energy transition is studied from different angles. However, no study systematically explores the influence of the availability of different alternative energy choices on the households' energy transition and their decision-making processes. Furthermore, no study specifically investigates the potential influence of the availability of different natural gas sources on energyrelated behaviours.

This gap is crucial as the availability of different RECs, technological developments, and climate change goals urges the households' energy

transition to take place faster. This is particularly highlighted in the context of the European households, considering the energy crisis resulting from the ongoing Russian-Ukrainian conflict, to reduce their natural gas consumption (and re-think about the sources of the natural gas imports) and eventually fasten the thermal energy transition [21]. However, the literature has not investigated the households' opinions towards all the alternative energy choices and to what extent the availability of different alternative energy choices (such as different natural gas from different countries/ origins) could potentially influence the (thermal) energy transition. For instance, the households should only have choices about the individual renewable energy systems that they adopt (e.g., individual heat pump), or they should be given choices on the municipality-level decision-making (e.g., collective renewable energy systems they can adopt) and national level decision-making (i.e., natural gas import sources). In the real world, this can be translated as to what extent public opinion should be involved in making national-level energy choices and how such choices could contribute to the energy transition as a whole. In other words, it is unclear whether the importance and potential of providing alternative energy choices for households at different levels could add to the energy transition.

To address this gap, this study aims to explain and describe the influence of the availability of alternative energy choices at different levels on individual decision-making to facilitate households' energy transition. To capture the decision-making processes and choices while including the heterogeneity of motives and behaviour of households, the research employs agent-based modelling and simulation (ABMS) instead of other simulation approaches such as System Dynamics [31], Discrete Event Simulations [32] and Equilibrium Modelling [33], that mainly focus on system processes and outcomes. Also, to address the ongoing energy crisis due to the Russian-Ukrainian conflict, the study focuses on the thermal energy choices of households within thermal energy communities (TEC initiatives) [17,34]. In this sense, the current study can be seen as an extension of existing models, such as [28,30,35], which are focused on establishing and functioning of TEC initiatives. Therefore, the model contributes to the existing literature by adding alternative energy choices at different levels to understand their influence on the individual's decision-making. The alternative energy choices at different levels considered in this study are: national level (i.e., sources of natural gas), municipal/ community level (i.e., collective renewable energy technologies, such as geothermal valves and biogas combined heat and power), and individual level (e.g., solar thermal and heat pump). Lastly, to address the research objectives concretely, this study uses input data from the Netherlands, a European country with unique characteristics for the thermal energy transition, as elaborated in [28,36]. Along with bringing structured insights, the developed model could potentially be used for other countries. To summarise, the study has the following contributions:

- Explaining and describing the influence of the availability of different alternative energy choices, particularly natural gas sources, on households' behaviour and their contribution to energy transition,
- Demonstrating the value of using (and expanding existing) ABMS to study new questions/ topics (e.g., different technical and policy choices),
- Providing concrete insights on thermal energy communities and thermal energy transition, particularly in Europe.

The study also aims to provide concrete insights and recommendations to relevant actors, particularly the policy-makers and individual households, along with scientific contributions. In addition, such insights and recommendations could contribute to the energy and geopolitical agendas at a higher level. More specifically, the study can be seen as a response to concerns in relation to the energy and geopolitical crisis in Europe and around the world.

The structure of the paper is as follows: Section 2 explains the theoretical background of this research. Section 3 elaborates on the

research approach, ABMS. Section 4 delineates the model conceptualisation. Section 5 describes the simulation results. Section 6 presents the discussions and limitations. Finally, Section 7 concludes the main findings and provides recommendations.

2. Theoretical background

This section introduces the theories and frameworks used as the backbone of the modelling exercise. Multi-level governance (MLG) is used to conceptualise and investigate different alternative energy choices. While the collective action perspective (i.e., the institutional analysis and development framework) is used to structure and understand the decision-making processes, the social value orientation theory is employed to structure individuals' heterogeneous behavioural attributes and motivations.

2.1. Multi-level governance for the Dutch public opinion in thermal energy transition

Multi-level governance (MLG) aims to capture and understand the political and institutional processes to describe the political processes and governance structures (such as the establishment of the European Union) [37,38]. The MLG structures such processes on different levels (e.g., global, continental, national and regional levels) from the perspective of different actors (e.g., public, government and businesses) and different sectors (e.g., energy, transport and agriculture) [37,38]. Various studies, such as [39-41], explored the MLG in the context of European integration and governance and expanded its applications. In recent years, the application of the MLG has expanded further and included climate and energy governance. For instance, [42] explored the challenges and uncertainty of natural resource management through the MLG, and [43] structured global climate governance by using the MLG. On the other hand, various studies such as [44–46] applied the MLG to study and facilitate energy transition. The deployment of solar energy for China's sustainable development processes and goals through the MLG is explored in [47]. Germany's energy efficiency policies are studied using the MLG in [48]. The multi-level governance of renewable energy in England is presented in [49]. Particularly at the local level, [45,46] studied the community energy systems through the MLG lens.

Following such studies, three levels of governance and policy and their representative choices are structured for this study:

- National level/ EU level- natural gas import sources and consumption: This level focuses on public opinion about the sources of natural gas available for individual household consumption. In a broad sense, there are three alternative choices: (i) the current mixture the natural gas grid offers, (ii) specifically choosing a natural gas from an ally country (in the case of the Netherlands, a European or North American country), or (iii) natural gas from a non-ally country (Russia and Qatar in the case of the Netherlands).
- Regional level- collective renewable thermal energy systems: This level focuses on the available renewable thermal energy systems the

city or province could use to cover the thermal energy demand. In the case of the Netherlands, three systems are prominent, namely: aqua thermal energy storage (ATES), biogas combined heat and power and electric boilers (i.e., electrification) [28,35].

• Individual level- individual renewable heating systems: This level focuses on the available renewable thermal energy systems for implementation within a single household to cover the thermal energy demand. Following studies such as [30,35], the following systems are suitable for the case of the Netherlands: heat pumps, small bio-energy heaters (i.e. wood pallets) and photovoltaic thermal hybrid solar collectors (i.e., Solar PVT)

Considering these three levels (and their representative choices) could potentially contribute to capturing the influence and application of multi-level governance and the availability of alternative energy choices for individual households. Therefore, the MLG is employed to structure the three levels of the decision-making processes and choices (i.e., individual, municipal and national levels) rather than focusing on investigating the governmental institutions. Furthermore, by employing and applying MLG and the three representative levels, such an approach contributes to studies such as [45] investigating the governance and policy processes for sustainability transitions.

2.2. Collective action: The institutional analysis and development (IAD) framework

As presented in [17,35], CESs can theoretically be seen as a form of collective action where the generation, distribution and consumption of renewable energy is the common goal of different actors (e.g., individual households). Therefore, by breaking a system down into simpler and manageable components, the institutional analysis and development (IAD) framework investigates the dynamics of the decision-making processes in such a system [50,51], as presented in Fig. 1.

The IAD framework investigates collective action problems from an institutional perspective, which are the rules of the game that govern the activities of the actors [52,53]. As a conceptual space, the action situation is the main component of the IAD framework, where actors make decisions while considering the trade-offs, take actions and experience the consequences [53]. The action situation component leads to a pattern of interactions that generate specific outcomes (e.g., establishment of TEC initiatives), which can be objectively assessed based on evaluation criteria (e.g., generated renewable energy and cost) [51]. A feedback loop connects the outcomes to the exogenous variables, ultimately influencing the action situation (and all the other components). The exogenous variables are classified into three main components: (i) biophysical conditions, which refer to the physical resources and capabilities available within the system's boundaries (e.g., available RETs); (ii) community attributes, include the cultural norms accepted by the community, shared values and beliefs (e.g., environmentally friendly behaviour); and (iii) rules-in-use; include the formal rules such as policies and regulations that officially govern the system (e.g., available subsidies and Carbon emissions tax) [30,53].



Fig. 1. The IAD framework [51].

The IAD has proven useful for building agent-based models, along with its analytical power and being highly instrumental in studying thermal energy communities (TEC initiatives) as a collective action problem [54]. Therefore, various studies used the IAD framework for conceptualising their experimental settings (e.g., [28,35,55]).

2.3. Individual behaviour: The social value orientation (SVO) theory

As presented in [56,57], individual households join an energy community to fulfil specific concerns (e.g. environmental) or to achieve certain goals (e.g. financial benefits). In this context, the social value orientation (SVO) theory explains the individuals' motivations, concerns and goals for making a specific decision, such as joining a CES [58]. SVO theory provides measurement tools as it hypothesises that people vary in their motives and goals when evaluating different resource allocations and alternative choices [59]. Therefore, SVO theory classifies individuals' personalities into four groups as follows:

- Altruistic: individuals are selfless, focusing on maximising joint benefits regardless of the impact on their payoff; the opportunity to help others is their motivation;
- Cooperative: individuals aim to maximise one another's outcome together with their own;
- Individualistic: individuals are mainly concerned with their outcomes, focusing on their payoff without having a specific motive for minimising another one's benefits;
- Competitive: individuals aim for maximum results and strive to minimise other individuals' benefits [59].

By such classification, SVO theory helps capture real-life decisionmaking situations while considering various motivations and goals [58]. As SVO theory is highly instrumental, various studies have employed it to study energy-related behaviours. [60,61] employed SVO theory to empirically explore the individuals' decision-making to adopt and invest in renewable energy technologies and energy efficiency. On the other hand, several studies used SVO theory as the theoretical backbone of their ABMS. For instance, [23] used SVO to explore the influence of behavioural attitudes and leadership on the deployment of CESs based on solar photovoltaic systems in the Netherlands. [30] used SVO theory to investigate the importance of the behavioural attributes towards collective energy security in TEC initiatives.

There are other frameworks which could potentially be useful. For instance, socio-ecological systems (SES framework) [62] could be used to delve into collective action and institutional analysis details. However, such a framework has been mainly applied to the (collective) systems that have already existed for some time rather than studying new and emerging systems and complex phenomena (such as thermal energy transition and TEC initiatives). The multi-level perspective (MLP) [63] could also be beneficial for studying the procedures and conceptualising TEC initiatives as niche innovations. However, the study focuses on the individual and collective decision-making processes for local energy transition rather than on dynamics for making TEC initiatives, as an established innovation, to become the dominating system for thermal energy generation and heat transition.

To summarise, the three elaborated theoretical concepts are used as a backbone of agent-based modelling and simulation (ABMS). Thermal energy communities (TEC initiatives) are seen as collective actions, and therefore, decision-making processes are conceptualised using the IAD framework. The SVO theory is employed to capture and categorise individual households' motivations and values for participating in TEC initiatives. The suggested multi-level governance approach (MLG) captures the different policy levels and their importance for establishing and functioning TEC initiatives. Using the mentioned theories as the backbone of ABMS serves the research aim, namely explaining and describing the influence of the availability of alternative energy choices on different levels for facilitating the households' energy transition.

3. Agent-based modelling as a computational simulation method

Performing real-world experiments would be time-consuming and costly, with not necessarily beneficial impacts on individuals' lives. However, computer simulation is often used to conduct experiments in a virtual simulation environment [64,65]. In this branch of literature, computational social simulation is a well-established field of research, integrating different scientific fields such as computer science, social sciences and technological design [66,67]. Agent-based modelling and simulation (ABMS) is specifically promising for this research as it facilitates the exploration of artificial societies of autonomous agents as representatives of the real world [68,69].

In an ABMS, "An agent is the software representation of some entity that completes an action or takes a decision, by which it effectively interacts with its environment" [70]. Therefore, agents are heterogeneous, autonomous and individual decision-making entities (such as individual households) that are able to learn and interact with each other and their environment [68,71]. In addition to studying and capturing individuals' behavioural choices, using ABMS provides the opportunity to study the emergent behaviour of the system [72]. Moreover, ABMS inherently adds the time dimension, which allows the examination of different scenarios under different input settings over time [69,72].

For these reasons, the investigative power is enhanced, and therefore, ABMS is considered a suitable approach for studying the dynamics and interactions within TEC initiatives to explain and describe the influence of the availability of alternative energy choices. Although considering the complexity of the real world, an ABMS cannot represent all the details of real-world decision-making processes, different studies argue for and use ABMS to study different topics in the context of energy transition. Studying value conflict for acceptance of decentralised energy systems [73], simulating behavioural attitudes [74] and leadership in the energy communities [23], exploring the social acceptance of sustainable heating systems [75], studying local heating systems [76,77], indoor heating and cooling and built environment systems [78-80], modelling and simulating zero energy communities [81-83], and studying the renewable energy technology adoption [84,85], renewable energy market design and price reforms [86,87], are examples of such studies in this branch of literature. A detailed overview of studies that employed ABMS for studying (thermal) energy communities is presented in [88].

4. Model conceptualisation

This section describes the model conceptualisation and implementation using the IAD framework, as presented in Section 2.2, starting with agents, followed by exogenous variables (i.e., biophysical conditions, attributes of community and rules in use), action situation and interactions, evaluation criteria and outcomes.

4.1. Agents

The model represents a municipality with multiple neighbourhoods, each one of which can only form and implement one thermal energy community (TEC initiatives). The following two types of agents are included in the model:

(i) Individual households who initially are connected to the national grid to cover their thermal demand by consuming natural gas. Following studies such as [30,35], individual households (i.e., agents) have the following internal motivations to participate in TEC initiatives: (i) energy independence, (ii) a sense of community, (iii) environmental concern and (iv) economic benefits, which independently from each other, each of them have a value between 0 and 10 (i.e., 0 is the weakest and 10 is the strongest). Based on their interaction, individual households in a neighbourhood can influence each other's internal motivation. Individual households make decisions related to participating in TEC

initiatives and adopting renewable heating systems based on their SVO type, which is captured based on these four internal motivations as following:

Level of motivation =(environmental concern + sense of community) -(financial concerns + energy independence)

- If Level of motivation >1: SVO-type 1 (i.e., altruistic),
- If Level of motivation < -1: SVO-type 3 (i.e., individualistic),
- If Level of motivation ≥ −1 and ≤ 1, and sense of community <5: SVO-type 4 (i.e., competitive),
- If Level of motivation ≥ -1 and ≤ 1 , and sense of community ≥ 5 : SVO-type 2 (i.e., cooperative).

In addition to [19], following studies such as [89–91], individual households also have two other internal motivations which potentially influence their connection and consumption from the national grid (in this case, natural gas consumption): national energy independence and awareness/care towards the energy source they consume. These two internal motivations will be used for the decision-making process described in Section 4.5. (the data is presented in Appendix E, from [21]). Finally, following [28], among all the individual household (i.e., agents in the model), the top five most environmentally friendly ones whose other motivations are also higher than the median value (i.e., \geq 5) are considered to be the community board, who are representatives of individual households in the neighbourhood and are responsible for initiating, deciding and leading the establishment and functioning of TEC initiatives.

(ii) The municipality represents the national government (in this case, the Dutch government; depending on the context, it could be any government), which is responsible for the thermal energy transition and supporting TEC initiatives that are being formed and implemented. In this modelling exercise, the municipality performs its responsibilities by checking the eligibility requirements for and distributing the available subsidies among the TEC initiatives. This is in line with municipalities' role in different (European) countries, as presented in [35,92]. As the municipalities' resources (such as available budget) are limited, following [28,35,93], municipalities have four strategies to perform, namely: environmentally driven (meaning supporting the TEC initiative with the most CO₂ reduction option), economically driven (meaning supporting the TEC initiative with the least economic burden for the municipality itself), socially driven (meaning supporting the TEC initiative with the most involved participants in a neighbourhood) and a trade-off between the three. Individual households have been aware of the municipality's strategy since the beginning of the simulation. The municipality strategy has been known for the individual households since the beginning of the simulation. In the real world, this means the municipality (and/or the national government) has transparently communicated the criteria for granting subsidies to TEC initiatives, and the individual households are aware of the criteria for granting subsidies TEC initiatives.

4.2. Biophysical conditions: Alternative energy choices

Following studies such as [35], the agents in the model (i.e., individual households) have the following alternative energy choices to select from (based on their internal values and interactions, as explained in detail in Section 4.5.):

- Adopting renewable thermal energy systems:
- o Collective renewable thermal energy systems: the choices are biogas heaters, aquifer thermal energy systems (ATES), and electric boilers (will be distributed through medium-temperature district heating);

- o Individual renewable thermal energy system: the choices are heat pumps, small bio-energy heaters (i.e. wood pallets) and photovoltaic thermal hybrid solar collectors (i.e., Solar PVT).
- Continuing consuming natural gas from:
 - o the current mixture of the natural gas grid (i.e., natural gas from different origins and countries) for the individual households who are indifferent to the natural gas source;
 - o an ally country (in the case of the Netherlands, a European or North American country) for the individual households who care about national energy independence;
 - o another country that is not an ally (e.g., Russia and Qatar in the case of the Netherlands) for individual households who are economically driven

4.3. Attributes of community

(1)

Each individual household lives in a neighbourhood with its own specific TEC initiative (meaning each individual household can only participate in one TEC initiative). Each month (i.e., each time step in the model represents a month), the households in a neighbourhood can interact with each other randomly in monthly residential meetings that are held for decision-making related to CES. These interactions can be translated as a form of information exchange, peer pressure and influence on each other in the real world. The small world network is used to capture and simulate such interactions [94]. The 'small world' is commonly used in modelling the interactions in neighbourhoods, particularly for studying CESs (e.g. [23,28,95]).

Such interactions lead to influencing each others' motivations in the following manner as presented in [30,96]: for each internal motivation (i.e. energy independence, trust, environmental concern, economic benefits, national energy independence and awareness/care), if the value is not extreme, between 2 and 8 (meaning not hard to be changed), they will be updated and leaning one value towards the interacting agent's opinion, for better or worse. If the value of internal motivations is higher than 8 or lower than 2 (meaning they are extreme and hard to change), the agent will not update its own values during the interaction and, therefore, will not be influenced.

4.4. Rules-in-use

Three types of rules are considered: (i) supportive policies such as subsidies, (ii) prohibiting policies such as CO_2 tax, and (iii) flexibility in natural gas import sources. The Dutch subsidy schemes such as 'Stimuleringsregeling Duurzame Energie' (SDE++) and reports from the Dutch Environmental Assessment Agency (PBL) are used to collect data related to the policies. The model assumes that sources are available for natural gas imports and that they can be used depending on households' choices. The current mixture of national natural gas grid data is extracted from PBL, Statistics Netherlands (CBS) and Eurostat.

4.5. Action situation and interactions

The households have a period to exchange information to know each others' motivations and align them. Individual households consume natural gas from the national gas grid during this information exchange period (and before deciding on alternative energy choices). These interactions are based on the description in Section 4.3. Decisions 2, 3, 4 and 5 (type and amount of collective and individual renewable thermal energy systems and the choice of natural gas consumption) are based on the MLG (see Section 2.1). After a period of information exchange among the individual households, the following decisions need to be made:

(i) The project leadership: to choose the project leader, between the community board or municipality, who is responsible for organising the TEC initiative formation and implementation; individual households check the municipality's strategy, which will be one of the following:

- Environmentally driven strategy: if the individual households' environmental concern is higher than 5 and belongs to first and second SVO-types (meaning being altruistic or cooperative), it votes for the community board leadership. If the environmental concern is higher than 5 and belongs to third and fourth SVO-types (meaning being individualistic or competitive), the individual household checks its "sense of community" value, which, if it is higher than 5, votes for the community board leadership.
- Societally driven strategy: the procedure is the same as that described for the environmentally driven strategy. The only difference is that the households first check the "sense of the community" value, and then if needed, they check the "environmental concern" value.
- Economically driven strategy: if the individual households' economic benefits are higher than 5 (regardless of its' SVO-type), it votes for the municipality leadership.
- Trade-off strategy: randomly, with an equal chance, the individual households opt for one of the above-mentioned strategies.

(ii) Type of the collective renewable thermal energy system: Depending on the project leadership, the decision-making process for the collective renewable thermal energy goes through two different processes for choosing between biogas heaters, ATES, and electric boilers as collective technologies.

- If the municipality is the leader, depending on its strategy (i.e., environmental, economic, social and trade-off), the municipality chooses one of the collective renewable thermal energy systems and then communicates its choice with the individual households. Therefore, it is more of a top-down approach. As presented in Appendices A and B, the municipality calculates and normalises values for each technology's annual CO2 emissions, costs and minimum participants to make this decision.
- If the community board is the leader, they first go through a multicriteria decision-making process (MCDM) to select a collective renewable thermal energy system based on the board members' internal motivation (and by following calculations presented in Appendices A, B, and D). The board suggests the technology with the highest MCDM score as the alternative thermal technology for the community. After this suggestion, each individual household conducts an individual MCDM based on its own internal motivations and values, and if the suggested technology fulfil the following conditions, it would be accepted: (i) the suggested technology is the one that is rated as the highest by more than half of the neighbourhood; (ii) the suggested technology by the community board is not the technology that is rated as the lowest by more than one-third of the neighbourhood. If the first technology proposed by the community board does not meet these conditions, the same procedure takes place for the next collective energy technology.

(iii) Capacity of the collective renewable thermal energy system: After choosing the type of the collective renewable thermal energy system (between biogas heaters, ATES, and electric boilers), individual households select the capacity of such a system, which is calculated in terms of the percentage of the total thermal demands of the individual households. The capacity is determined as the average percentage value of all the individual households, corresponding to the share of each individual household's thermal demand covered by the collective renewable thermal energy. The two conditions for this choice are the budget and SVO type of an individual household. The upper limit is how much an individual household can afford to invest in the collective renewable thermal energy within its budget. If the household belongs to the SVOtype 1 (i.e., altruistic), it prefers to meet all its demands collectively (i. e. 100%). For the other SVO-types, the preferences to cover their energy demand collectively is as follows: Households with SVO-type 2 (i.e., cooperative) 90%, households with SVO-type 3 (i.e. individualistic)

80%, and households with SVO-type 4 (i.e. competitive) 70%. If the collective renewable thermal energy does not cover the entire demand within the TEC initiative, the individuals, depending on their internal motivations, have other choices to address their thermal demand.

(iv) Type and capacity of individual renewable thermal energy system: after selecting a collective renewable thermal energy system, if the individual households' environmental concerns are higher than economic benefits, it goes through the following processes:

- If its budget is sufficient, it should adopt one of the individual renewable thermal energy systems (between heat pump, wood pellet, and solar PVT) to cover its remaining thermal energy demand.
- If its budget is insufficient, it chooses to face blackout/ discomfort voluntarily; this means it reduces its demand and, therefore, saves money to install individual renewable thermal energy systems in the future.

When the individual household equally values environmental concerns and economic benefits, the sense of community value serves as a tiebreaker, meaning if the sense of community is lower than 5 the household withdraws from adopting collective and individual renewable thermal energy systems.

(v) Choice of natural gas consumption: after selecting a collective renewable thermal energy system, if an individual household has economic benefits higher than environmental concerns, it chooses to use natural gas as the energy source for the remaining demand that is not covered by the selected collective heating energy system and goes through the following processes:

- If it is less concerned/ aware of the natural gas sources (i.e., does not care about the source of natural gas), it will continue with what the current natural gas grid offers (meaning the current equal mixture of natural gas from ally countries and non-ally countries collected from CBS and Eurostat; 50% from ally countries and 50% from non-ally countries).
- If it is highly concerned/ aware of the natural gas sources (i.e., cares about the source of the natural gas) and has higher national energy independence concerns than economic benefit, it selects to get the natural gas from its own country (in this case, the Netherlands). If its country does not have natural gas resources, it chooses to get it from an ally country (in this case, a European or North American country). If this were also impossible (the allied country does not have natural gas), the individual household would voluntarily face a blackout/discomfort, reducing its demand and saving money for the future. Depending on its interactions in the next ticks (see Section 4.3), the individual household might grow towards investing its savings in (collective and individual) renewable thermal energy systems.

(vi) Financing, establishing and maintaining TEC initiatives: after selecting the technical configurations (i.e., natural gas sources, collective and individual renewable thermal energy systems), the final financial feasibility of the system, including applying for the subsidies, takes place. In this phase, the project leader (either the municipality or community board) calculates the renewable generation (including collective and individual technologies), CO₂ emissions, number of adopters of RETs and costs. Then, the subsidy amount is calculated. Every 12 ticks in the simulation (meaning once a year), the municipality considers all the subsidy applications and ranks them based on its own strategy. The municipality distributes the subsidy from the top of the list until its subsidy is finished. If a TEC does not receive the subsidy (as it might not meet the municipality's criteria for receiving the subsidy or as it might be low in the municipality's ranking while the subsidy is finished), it waits for the next year and applies again.

After the investment is fulfilled (including receiving a subsidy and individual households' investment), the construction of the infrastructure and energy systems will take one year (12 ticks in the simulation).



Fig. 2. Conceptual framework of model flowchart.

After establishing, every year, the project leader checks whether they have reached the end of their project time horizon (i.e., 20 years in the simulation). As the agents continue to interact with each other and influence each others' values during the simulation, "non-adopters" in the neighbourhood (i.e., non-members of TEC initiatives) can re-evaluate their participation and can potentially add to the community initiative over time. Fig. 2 presents these six main decision-making processes as a conceptual model flowchart.

4.6. Evaluation criteria and outcomes

The evaluation criteria and outcomes are considered key performance indicators (KPIs) in the simulation. In order to explain and describe the importance of the availability of choices for individual households, along with analysing the alternative energy choices (including the technical configurations and average natural gas consumption), the following KPIs are used (and their calculations are presented in Appendix C):

- Average voluntary blackout/ discomfort per household (%): this KPI is the average voluntary blackout/ discomfort for individual households. As presented in Section 4.5., an individual household might face voluntary blackout/ discomfort due to technical, economic and behavioural conditions.
- Average CO₂ emission per household per year (KgCO₂/year): the average CO₂ emission per year of an individual household is based on the CO₂ intensity of the selected energy choices.
- The average costs per household (€/month): The average cost per year for each individual household is calculated based on three energy costs: collective renewable thermal energy system, individual renewable thermal energy system and natural gas consumption.

4.7. Parameters and experiment setting

For capturing the Dutch individual households' internal motivations, the results from [19] are used. Table 1 delineates the model's parameter configurations based on these parameters and the one-factor-at-a-time (OFAT) sensitivity analysis approach [97]. These are also in line with the previous studies, such as [28,30,35], that studied the TEC initiatives in the Dutch context.

As the goal of this study is to explain and describe the importance of the availability of choices for individual households (and TEC initiatives) at different decision-making levels (rather than focusing on understanding technical and social conditions), the model adopted its parameters and experimentation settings from relevant studies such as [28,30,35], which are proven to be useful and instrumental for presenting the Dutch complex thermal energy system. Such parameter setting could facilitate comparing the results with the previous studies where the possibility of choosing natural gas sources is not included as an alternative energy choice. The parameters presented in Table 2 are used to conduct the simulation, which led to 48,600 simulated TEC initiatives (and 29,160,000 individual households in total).

Table 1

Input parameter configuration settings.

Parameter	Value	Unit
Duration of information exchange	7	Months
Neighbourhood size	600	households
Number of connections each household has	3	Number
Number of neighbourhoods in a municipality	3	Neighbourhood

Table	2
-------	---

Experimentation settings.

Parameter	Value	Unit
Increasing rate of the natural gas price	0.01, 0.02, 0.03	(€/kWh)
CO ₂ taxes	0.002, 0.004, 0.006	(€/kg)
Ambient temperature changes	Mild, High, Severe	_
Available subsidy	2, 4, 6	Million €
Municipality subsidy policy	Environment, social, economic, a trade-off	-

5. Results

This section presents the simulation results in two main categories: (i) an overview of the alternative energy choices and technical configurations and (ii) an overview of environmental, costs and discomfort indicators.

5.1. Overview of alternative energy choices and technical configurations

Fig. 3 demonstrates that individual households who joined thermal energy communities (TEC initiatives) adopted collective and individual renewable thermal energy systems, and overall, the simulated TEC initiatives drastically reduced their natural gas consumption. Few TEC initiatives continue consuming around 30% of their thermal demand from natural gas, while, on average, including all the 48,600 simulated TEC initiatives, the average natural gas consumption is 12.6% per initiative. These results are considerably lower than those of the previous study (where the individual households/ agents did not have the opportunity to choose their natural gas import source), as presented in [30,35]. In these studies, the alternative energy choices are only on two levels (i.e., collective and individual renewable thermal energy systems), and the average TEC initiatives natural gas consumption is 37% in [30] and around 20% in [35]. As the main difference between the studies is related to the possibility of choosing natural gas sources, this demonstrates the importance of giving individual households such a choice. In other words, the possibility of choosing the source of natural gas could lead to consumption reduction and, eventually, its imports. Overall, these results could translate to a higher contribution of TEC initiatives in the energy transition and reducing CO₂ emissions.

In the same line, as Fig. 4 presents, considering all the 48,600 simulated TEC initiatives, the average generation for collective thermal energy system is 83%, which is also higher than previous studies. This indicates that the simulated individual households prefer to get their thermal energy from collective renewable thermal energy systems rather than a foreign country that does not align with their internal motivations.

Fig. 5 demonstrates the alternative energy choices in more detail,

 $[3 \times 3 \times 3 \times 3 \times 4]$ (permutation of paremeters' combination) $\times [50]$ (reputation of each combination) $\times [3]$ (number of neighbourhoods in each simulation) = 48600 (simulated TEC initatives in total)

(2)

48600 (simulated TEC initiatives in total) \times 600 (individual households in each of TEC initiatives) = 29 160 000 (simulated individual households in total) (3)



Fig. 3. Average natural gas consumption.



Fig. 4. Percentages of collective thermal energy generation.

divided by collective and individual thermal energy systems and natural gas consumption (from allies and other countries). As the technical and institutional settings for collective thermal energy systems in this study are similar to those in the previous study, as elaborated in Sections 1 and
 Table 3

 Details of alternative energy choices in percentage.

		Collective thermal energy systems			Total (%)
		ATES	Biogas	Electric boiler	-
Individual energy	Heat pump	40.7	14.9	5.0	60.6
choices	Ally gas	5.2	2.2	0.7	8.1
	Other gas	4.7	2.0	0.7	7.4
	Solar	0.2	0.0	0.0	0.2
	Wood pellet	0.0	0.0	0.0	0.0
	Solar waiting	0.1	0.0	0.0	0.1
	Wood pellet waiting	0.0	0.0	0.0	0.0
	Heat pump waiting	15.7	5.8	2.1	23.6
Total (%)		66.6	24.9	8.5	100%

4.7, the distribution of the collective thermal energy systems stays mostly the same in comparison with [35]. ATES and heat pumps are the most chosen thermal energy systems as they have relatively better environmentally friendly and economical performances. However, as presented previously, the capacity of such collective technologies has increased (as presented in Fig. 4).

Compared with previous studies such as [25], Fig. 5 clearly shows that households (i.e., agents in the simulation) replace their natural gas consumption with individual heat pumps. The details of the percentages are presented in Table 3.

As Fig. 5 and Table 3 illustrate, heat pumps dominate individual renewable energy choices. Most communities (approximately 60% of 29,160,000 simulated individual households in 48,600 TEC initiatives) choose heat pumps as their individual renewable thermal energy system. On top of that, approximately 23% of individual households choose temporary blackouts to install heat pumps in the near future. These households are either (i) the ones who are environmentally friendly but do not have enough money for the heat pump immediately or (ii) the ones who have a high value for the energy independence of their country but do not have the money to invest in heat pump immediately. On the other hand, the wood pellet and solar were not adopted by households and in TEC initiatives (only 0.3% solar in total), which could be potentially related to their economic and environmental performances. As Fig. 5 and Table 3 show, natural gas from allies (e.g., Norway and the UK) has almost the same share as natural gas from other sources (e.g.,



Distribution of collective and individual energy source combinations

Fig. 5. Distribution of energy alternative choices in all runs.

J. Fouladvand et al.

Table 4

Overview	of	KPIs.
----------	----	-------

	Minimum	Average	Maximum
Voluntary blackout/ discomfort (%)	0,3	2,7	8,4
Average costs per month (€/month)	90,89	128,41	230,85
Average CO ₂ emission per year (KgCO ₂ / month)	131,9	151,1	173,8

Qatar). All these could be translated as the availability of more alternative energy choices, potentially leading to a more sustainable choice, in this case, collective and individual renewable thermal energy systems.

5.2. Overview of other KPIs

As mentioned in Section 4.6., along with analysing the alternative energy choices and technical configurations, three other KPIs are also measured in the simulation presented in Table 4.

As Table 4 presents, considering all 48,600 simulated TEC initiatives, the average voluntary blackout/ discomfort is 2.7%, which is considerably lower than the previous studies (e.g., [29,30,35]). Although there are various possibilities that individual households could face blackout/ discomfort (as explained in Section 4 and Fig. 2), as households have alternative energy choices, they could still manage better avoiding it. However, this resulted in increasing the costs for individual households, as could be expected. Compared to studies such as [18,20], the average costs are approximately 20–25% higher. Increasing the share of (collective and individual) thermal energy systems and reducing natural gas consumption also significantly reduce CO_2 emissions.

6. Discussion and limitations

By capturing and simulating the Dutch thermal energy transition, with its unique characteristics and factors, the study illustrated the importance of the availability of alternative energy choices for the (local) energy transition. In the modelling exercise, the individual households (i.e., agents in the simulation) had three layers of alternative energy choices: (i) natural gas (i.e., the current mixture of the grid, ally and other countries), (ii) collective renewable thermal energy systems (i. e., ATES, biogas and electrification), and (iii) individual renewable thermal energy systems (i.e., heat pump, wood pellet, solar PVT).

6.1. Discussion

The main difference between this study and similar ones in the field (e.g., [28,30,35]), and eventually one of its contributions, is related to adding the decision-making loop on the natural gas source. In other words, the previous studies included only the two lower levels of multilevel governance (MLG) (e.g., individual and municipal decision-making levels), and the possibility of choosing a natural gas source was not investigated. Although the real-world application of such findings is challenging (i.e., each household chooses its own sources of natural gas), this could be translated into policies on a higher level (e.g., including public opinion on the sources of natural gas). The study showed that the availability of alternative energy choices (i.e., including the choice of the source of natural gas) could amplify the contribution of individual households and thermal energy communities (i.e., TEC initiatives) to the local energy transition. By employing multi-level governance (MLG), the study explored and explained the need to include public opinion in all three mentioned levels of decision-making in governmental institutions. These insights are in line with the findings of previous empirical studies such as [20,26,98], which emphasise the importance of institutional settings (such as including public opinion and developing supportive policies) for the local (heat) transition.

The simulated TEC Initiatives drastically reduced their natural gas consumption, on average, by approximately 20%. In order to cover the

thermal energy demand without natural gas consumption, the TEC initiatives mainly increased the capacity of their collective renewable thermal energy systems (25%-30% increase compared to previous studies, where the possibility of choosing natural gas sources is not included) or adopted more individual heat pumps (15%-20% increase compared to previous studies). Such choices and configurations led to a significant decrease in individual households' CO₂ emissions (decreased by 90% approximately compared to using fully natural gas for thermal energy demand) and reduced the voluntary blackouts/ discomfort that individual households face (on average 2.7% compared to previous studies it 5%-10% decreased).

Such improvements are reflected in individual households' average costs, which resulted in an average of 128,41 (ϵ /month) over 20 years of simulation. This is almost two times higher than using a thermal energy system based on fully natural gas, in comparison with energy prices from 2020. It is approximately 30%–35% more expensive than previous studies that studied TEC initiatives (e.g., [28,35]). Financial feasibility is one of the most important factors that influence the deployment of innovative systems (as extensively discussed in [98]), and therefore, to make such multi-level collective action complex systems feasible, it needs to be addressed. Shifting the available subsidies from fossil fuels to RETs, new business strategies (examples are presented in [99,100]) and the CO₂ emissions tax (as it is currently implemented for Dutch companies and as presented in policy reports), are the examples of economic solutions for making the suggested systems more financially attractive and feasible.

The study showed the importance of public opinion on the higher level of governance (e.g., natural gas imports) and the performance of TEC initiatives. These insights contribute to studies such as [21,22,25,26], which empirically investigated the different factors influencing the local heat transition and natural gas consumption reduction. The findings could also contribute to establishing and functioning other types of energy communities, such as solar-based energy communities, as there are some similarities between the challenges, as presented in [101]. Considering all these points, it is concluded that the availability of alternative energy choices on different levels (e.g., national, municipal and individual levels) could potentially lead to more desirable (i.e., environmentally friendly and sustainable) outcomes, in this case, by amplifying the contribution of TEC initiatives to the energy transition.

6.2. Limitations and future research

Although this study brought new insights into the sustainable (energy) transition, energy policy and institutional design, it has certain limitations.

Firstly, ABMS as a method has certain limitations. Although the modelling practice and its results provided new insights, ABMS represents a simplified version of reality like other computational modelling practices. The real world is more complicated, and other factors, such as fluctuations in natural gas prices and introducing of new energy policies, could potentially influence the outcomes. Therefore, considering new research questions and aims, the future models could potentially include such factors to understand and explore them within the context of thermal energy transition and TEC initiatives. Furthermore, to address and capture such complexities, different research methods such as optimisation modelling (to explore the energy system from a top-down perspective), and equilibrium modelling (to explore the supply-demand dynamics in a CESs) could be beneficial.

Secondly, like any other modelling exercise, the study and the model have specific assumptions that are simplified for their purpose. For instance, public opinion towards natural gas only had three choices: a mixture of the grid, ally countries, or other countries, which does not make a distinguish between the countries in the same category (the ally countries are used as an overarching category, and do not get to the details of each country in this category). Or the model did not include all the possible energy sources (e.g., hydrogen). Although such assumptions serve the aim of the study and provide insights related to the importance of public opinion and the need for the availability of alternative energy choices, they could be modelled in more detail in the follow-up studies (e.g., collecting data and including specific countries as natural gas sources).

Thirdly, to provide meaningful and concrete insights, the study focused on the Dutch thermal energy transition as its case study and heavily relied on comparing its results with previously published models (as the earlier versions of the described model). The choice of the case study (and all related assumptions) is justified; however, in future research, populating the current models with data from other countries (e.g., Germany and Denmark) could be beneficial in verifying the current findings further and bringing new insights. Exploring a new case study could also potentially lead to the expansion and development of the current version of the model further to capture new technical and institutional settings (e.g., renewable energy technologies and subsidy schemes) and actors (e.g., business owners).

Lastly, although the theoretical backbone of the model, the SVO theory, the IAD framework and the multi-level governance (MLG) were instrumental and novel in achieving the study's aim, they have certain limitations, as they frame the thermal energy transition in a certain way. However, applying theories such as the Theory of Planned Behaviour (TPB) [102] and The Behavioural Reasoning Theory (BRT) [103] could have led to more detailed insights regarding the individuals' concerns, motivations and values in the decision-making processes. The four-layer model of Williamson [104] could also be used to understand the feedback loops and dynamics of the development of institutional arrangements. Using the multi-level perspective (MLP) [63] could also lead to conceptualising TEC initiatives as niche innovations and understanding the dynamics of the deployment, making TEC initiatives become the dominating energy systems (i.e., socio-technical landscape) for providing thermal energy for households.

7. Conclusions and recommendations

As one of the main sources of greenhouse gas (GHG) emissions, the energy sector is going through a fundamental transformation called the energy transition. In this context, special attention is being placed on the local level, as well as on household roles and contributions to the energy transition. This research aimed to explain and describe the influence of the availability of alternative energy choices for facilitating the households' (thermal) energy transition. Thermal energy communities (TEC initiatives), as local and collective energy systems for generating, distributing and consuming renewable thermal energy for individual households, are chosen to be the underlying complex energy system that this study investigated. In this context, agent-based modelling and simulation (ABMS) with the SVO theory, the institutional analysis and development (IAD) framework and the multi-level governance (MLG) as its theoretical backbone are employed and populated based on the available data from the Netherlands. Within this modelling exercise, the agents (i.e., households) had three layers of alternative energy choices: (i) national level: natural gas sources (i.e., the mixture of the grid, ally and other countries), (ii) municipality level: collective renewable thermal energy systems (i.e., ATES, biogas and electrification), and (iii) individual level: individual renewable thermal energy systems (i.e., heat pump, wood pellet, solar PVT).

The results demonstrate the significant importance and influence of the availability of different alternative energy choices on the performance of TEC initiatives, such as reducing the natural gas consumption to 12.6% on average for all 48,600 total simulated TEC initiatives. Individual households increased the share of collective renewable energy generation (on average 83% for all TEC initiatives) and adopted more heat pumps (84% of all households in TEC initiatives) to meet their thermal energy demand. Such collective action, with vast choices on different levels, led to smaller voluntary blackouts/ discomfort (2.7% on average) while significantly contributing to reducing the CO_2 emissions of TEC initiatives (by 85%). On the other hand, it increased the costs of establishing and functioning TEC initiatives. Considering that including the choice of natural gas sources is the main difference of this study with the previous ones (e.g., [28–30,35,75,76]), results demonstrate the positive impact of availability of such choices on environmentally friendly outcomes (e.g., reducing natural gas consumption and reducing CO_2 emissions) for the local (thermal) energy transition. The availability of such choices can be translated into public opinion on natural gas resources; therefore, the study constructively delineated the influence of and the need to include public opinion in such high-level energy governance and policy-making processes.

7.1. Recommendations

The results from previous sections are translated into detailed recommendations as follows:

- Policy-makers are advised to ask for and include public opinion on high-level energy governance (e.g., sources of natural gas imports), as it could considerably contribute to reducing natural gas consumption and CO₂ emissions.
- Municipalities and policy-makers are urged to provide alternative energy choices for individual households and empower them in the related decision-making processes to accelerate the (thermal) energy transition.
- Individual households are encouraged to discuss and share their opinion towards alternative energy choices and the energy transition as a whole to provide input into the decision-making processes at the higher levels.
- All stakeholders are recommended to support and facilitate the establishment and functioning of (thermal) energy communities, as their contribution to the reduction of natural gas consumption and thermal energy transition is significant.

In addition to the concrete energy-related insights on the technical, institutional, and behavioural settings, the study also contributes to the computer modelling and simulation field, particularly computational social simulation, by (i) using multi-level governance as its backbone for the first time, and (ii) concretely demonstrating an example of comparing with (and expanding) existing models to explore a new question and provide new insights. Therefore, the study could be used by energy system modellers as an example of expanding their computational models to address new questions.

CRediT authorship contribution statement

Javanshir Fouladvand: Writing – original draft, Visualization, Validation, Software, Methodology, Investigation, Funding acquisition, Formal analysis, Conceptualization. Emre Ateş: Writing – review & editing, Visualization, Software, Methodology, Formal analysis. Yasin Sarı: Writing – review & editing, Validation, Software, Formal analysis, Conceptualization. Özge Okur: Writing – review & editing, Validation, Methodology, Investigation, Funding acquisition, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgements

The author would like to thank the European Commission for their financial support (project number: 101075587- SKILLBILL). Also, the

work received substantial support from the Energy Transition Lab of Technology, Policy and Management faculty of TU Delft. In addition, the support of Martin Junginger and Jesus Rosales Carreon for this study was highly appreciated.

Appendix A. Input data

Table 5

Data on collective thermal energy systems (the peak demand is considered 10% for all of them).

Variable	Units	Bioenergy	ATES	Electric boiler
Average capacity	kW	950	800	400
Capex	euros/kW	825	1600	800
Opex fixed	euros/kW/yr	55	113	120
Opex variable	euros/kWh	0.003	0.0019	0.025
Load hours	h/yr	3000	3500	2000
CO ₂ emission	kg/kWh	0.26	0.152	0.14
Lifetime	yr	20	30	30

Table 6

Data on individual thermal energy systems.

Variable	Units	Heatpump	Solar PVT	Wood pellet
Capex	euros/kW	1770	1450	415
Opex	euros/kW/yr	35.4	11	140
Load hours	h/yr	1500	700	2000
CO ₂ emission	kg/kWh	0.14	0.086	0.35
Lifetime	yr	15	20	20

Table 7

Other data (for the Netherlands).

Variable	Units	
CO ₂ intensity of electricity consumption for electric boiler	kgCO ₂ / KWh	0.43
CO ₂ intensity of electricity consumption for heat pump	kgCO ₂ / KWh	0.14
CO ₂ intensity of electricity consumption for solar PVT	kgCO ₂ / KWh	0.09
Average thermal energy demand per year	kWh	12,000
Gas price	euros/kWh	0.1
CO ₂ tax	euros/kg CO ₂	0.025
CO ₂ emission of natural gas	kg/kWh	0.2

Appendix B. Overall calculations of the leader

The leadership (either municipality or community board) calculations for choosing a collective renewable thermal energy system:

Fotal demand per year $=$ number of households $ imes$ average demand per household per year		
Annual CO2 emission = $[total demand per year imes CO2.intensity]$	(5)	
$\textit{Costs} (\textit{investment}) = \textit{Technology} \ \textit{capacity} \times \textit{Capex} + \textit{heat} \ \textit{demand} \times \textit{Operating} \ \textit{costs} \times \textit{lifetime}$		
Min.needed participants = $<\frac{Costs}{natural gas prices \times current consumption}$	(7)	

These values are then normalised on a scale between 0 and 1, where 0 represents the worst-performing alternative (i.e., highest emission, highest costs, or least number of needed participants) and 1 stands for the best-performing one. Then, the municipality ranks the technologies according to their normalised values and strategy (lowest emission first for environmental, lowest cost first for the economic and lowest number of participants for social).

Appendix C. KPIs calculations

C.1. Average voluntary blackout/ discomfort per household



Appendix D. Data for attributes of community

Table 8

MCDM inputs.

Criteria	Sub-criteria	Unit	Description	Reference
Financial criteria	CAPEX	e	Investment costs	[105]
	OPEX	€	Operational and maintenance costs during the lifetime of the system	[106]
	Payback time	Years	Years for the investment and maintenance cost to equal the accumulated energy savings from the change	[107]
	Subsidy coverage	%	Percentage of the capital costs covered by the subsidy (in the present study, this would be the SDE++ subsidy)	[106]
Environmental criteria	CO ₂ emissions	Kg CO ₂ eq	The CO ₂ emission intensity of technology based on capacity	[108]
	Land use	HA	Amount of land use required for technology based on capacity	[105]
	Social acceptance	1 to 10	The degree to which that technology is accepted, recognised and implemented	[106]
Independence criteria	The energy input to the system	kWh	Amount of energy input required for the technology to produce the heat to cover the neighbourhood heat demand	[108]

Appendix E. Data on energy independence and awareness/ care

Adopted from [21]	Likert scale 1 to 5, 1 meaning least agreeing with the statement, 5 strongly agreeing with the statement.				
Statements	Minimum	Maximum	Mean	Std Division	Variance
It is important for me to know the country's source of natural gas that I consume to heat my house. It is important for me to consume natural gas extracted from my country of residence.		5.00 5.00	3.49 2.69	1.10 1.04	1.21 1.08

References

- Masson-Delmotte V, Portner HO, Roberts D. IPCC Global warming of 15 C 2018; no. 9. https://doi.org/10.1017/CBO9781107415324.004.
- [2] Young J, Brans M. Analysis of factors affecting a shift in a local energy system towards 100% renewable energy community. J Clean Prod 2017;169:117–24. https://doi.org/10.1016/j.jclepro.2017.08.023.
- [3] Majcen D, Itard LCM, Visscher H. Theoretical vs. actual energy consumption of labelled dwellings in the Netherlands: Discrepancies and policy implications. Energy Policy 2013;54:125–36. https://doi.org/10.1016/j.enpol.2012.11.008.
- [4] Naus J, Van Vliet BJM, Hendriksen A. Households as change agents in a Dutch smart energy transition: on power, privacy and participation. Energy Res Soc Sci 2015;9:125–36. https://doi.org/10.1016/j.erss.2015.08.025.
- [5] Selvakkumaran S, Ahlgren EO. "Determining the factors of household energy transitions: A multi-domain study," *Technol. Soc.*, vol. 57, no. November 2018, pp. 54–75. 2019. https://doi.org/10.1016/j.techsoc.2018.12.003.
- [6] Irfan M, Zhao Z, Li H, Rehman A. "The influence of consumers ' intention factors on willingness to pay for renewable energy : a structural equation modeling approach," pp. 21747–21761. 2020.
- [7] Perlaviciute G, Steg L. The influence of values on evaluations of energy alternatives. Renew Energy 2015. https://doi.org/10.1016/j. renene.2014.12.020.
- [8] Frederiks ER, Stenner K, Hobman EV. Household energy use: applying behavioural economics to understand consumer decision-making and behaviour. Renew Sust Energ Rev 2015. https://doi.org/10.1016/j.rser.2014.09.026.
- [9] Dagili R. Do general pro-environmental behaviour, attitude, and knowledge contribute to energy savings and climate change mitigation in the residential sector ?193; 2020. https://doi.org/10.1016/j.energy.2019.116784.
- [10] Derkenbaeva E, Hofstede GJ, van Leeuwen E, Halleck Vega S. Simulating households' energy transition in Amsterdam: an agent-based modeling approach. Energy Convers Manag 2023;294, no. June:117566. https://doi.org/10.1016/j. enconman.2023.117566.
- [11] Nair G, Gustavsson L, Mahapatra K. Factors influencing energy efficiency investments in existing Swedish residential buildings. Energy Policy 2010;38(6): 2956–63. https://doi.org/10.1016/j.enpol.2010.01.033.
- [12] Santin OG. Behavioural patterns and user profiles related to energy consumption for heating. Energ Buildings 2011. https://doi.org/10.1016/j. enbuild.2011.06.024.
- [13] Ghofrani A, Zaidan E, Abulibdeh A. Simulation and impact analysis of behavioral and socioeconomic dimensions of energy consumption. Energy 2022;240:122502. https://doi.org/10.1016/j.energy.2021.122502.
- [14] Londo M, Matton R, Usmani O, Van Klaveren M, Tigchelaar C, Brunsting S. Alternatives for current net metering policy for solar PV in the Netherlands: A comparison of impacts on business case and purchasing behaviour of private homeowners, and on governmental costs. Renew Energy 2020;147:903–15. https://doi.org/10.1016/j.renene.2019.09.062.
- [15] Rogulj I, Peretto M, Oikonomou V, Ebrahimigharehbaghi S, Tourkolias C. Decarbonisation policies in the residential sector and energy poverty: mitigation strategies and impacts in central and southern Eastern Europe. Energies 2023;16 (14):1–21. https://doi.org/10.3390/en16145443.
- [16] Amores AF, Maier S, Ricci M. Taxing household energy consumption in the EU: the tax burden and its redistributive effect. Energy Policy 2023;182, no. August: 113721. https://doi.org/10.1016/j.enpol.2023.113721.
- [17] Fouladvand J. Thermal energy communities: what, why and how to formulate complex collective action for the thermal energy transition in Europe. Environ Res Lett 2023. https://doi.org/10.1088/1748-9326/acdd14.
- [18] Walker G, Devine-Wright P. Community renewable energy: what should it mean? Energy Policy 2008;36(2):497–500. https://doi.org/10.1016/j. enpol.2007.10.019.
- [19] Koirala BP, Araghi Y, Kroesen M, Ghorbani A, Hakvoort RA, Herder PM. Trust, awareness, and independence: insights from a socio-psychological factor analysis of citizen knowledge and participation in community energy systems. Energy Res Soc Sci 2018;38(January):33–40. https://doi.org/10.1016/j.erss.2018.01.009.
- [20] Sciullo A, et al. Exploring institutional and socio-economic settings for the development of energy communities in Europe. Energies 2022;15(4):1–22. https://doi.org/10.3390/en15041597.
- [21] Fouladvand J, Fiori F. Heliyon Perception towards reducing natural gas consumption and imports in Europe : A theoretical and empirical investigationvol. 10, no. April; 2024. https://doi.org/10.1016/j.heliyon.2024. e30719.

- [22] Okur Ö, Fiori F, Fouladvand J. Adoption of renewable heating systems and thermal energy communities in the Netherlands : an empirical study. Energy Rep 2024;11(January):3815–23. https://doi.org/10.1016/j.egyr.2024.03.036.
- [23] Ghorbani A, Nascimento L, Filatova T. Energy Research & Social Science Growing community energy initiatives from the bottom up : Simulating the role of behavioural attitudes and leadership in the Netherlands. *Energy Res* Soc Sci 2020; 70, no. March:101782. https://doi.org/10.1016/j.erss.2020.101782.
- [24] Hartmann K, Palm J. The role of thermal energy communities in Germany's heating transition. Front Sustain Cities 2023;4. https://doi.org/10.3389/ frsc.2022.1027148.
- [25] Teladia A, van der Windt H. Citizen participation gaps and challenges in the heating transition: learning from Dutch community initiatives. Renew Sust Energ Rev 2024;189, no. PA:113975. https://doi.org/10.1016/j.rser.2023.113975.
- [26] van der Schoor T, van der Windt HJ. Negotiating Dutch citizen-led district heating projects: Managing internal, external, and material networks to achieve successful implementation. Energy Res. Soc. Sci 2022;102(July):103166. 2023, https://doi.org/10.1016/j.erss.2023.103166.
- [27] Nava-Guerrero Del GC, Hansen HH, Korevaar G, Lukszo Z. The effect of group decisions in heat transitions: an agent-based approach. Energy Policy 2021;156, no. April:112306. https://doi.org/10.1016/j.enpol.2021.112306.
- [28] Fouladvand J, Aranguren M, Hoppe T, Ghorbani A. Simulating thermal energy community formation : institutional enablers outplaying technological choice. Appl Energy 2021;no. xxxx:117897. https://doi.org/10.1016/j. apenergy.2021.117897.
- [29] Fouladvand J, Verkerk D, Nikolic I, Ghorbani A. Modelling energy security: the case of Dutch urban energy communities, no. Cml Springer International Publishing 2022. https://doi.org/10.1007/978-3-030-92843-8_30.
- [30] Fouladvand J. Behavioural attributes towards collective energy security in thermal energy communities: environmental-friendly behaviour matters. Energy 2022;261, no. PB:125353. https://doi.org/10.1016/j.energy.2022.125353.
- [31] Ouyang M. Review on modeling and simulation of interdependent critical infrastructure systems. Reliab Eng Syst Saf 2014;121:43–60. https://doi.org/ 10.1016/j.ress.2013.06.040.
- [32] Edition S. Introduction to Discrete Event Systems Introduction to Discrete Event Systems. 2024.
- [33] B DP, J JW, editors. Handbook of computable general equilibrium modeling. vol. 1. Elsevier; 2013 [Online]. Available: https://www.scopus.com/inward/record. uri?eid=2-s2.0-84975884335&partnerID=40&md5=3fa82bebd7b3678338622 1e75386633d.
- [34] Fouladvand J, Ghorbani A, Mouter N, Herder P. Analysing community-based initiatives for heating and cooling: a systematic and critical review. Energy Res Soc Sci 2022;88:102507. https://doi.org/10.1016/j.erss.2022.102507.
- [35] Fouladvand J, Ghorbani A, Sari Y, Hoppe T, Kunneke R. Energy security in community energy systems : an agent-based modelling approach. J Clean Prod 2022;366, no. May:132765. https://doi.org/10.1016/j.jclepro.2022.132765.
- [36] Niessink R, Rösler H. Developments of Heat Distribution Networks in the Netherlandsno. June; 2015 [Online]. Available, ftp://ftp.ecn.nl/pub/www/l ibrary/report/2015/e15069.pdf.
- [37] Scharpf FW. Introduction: the problemsolving capacity of multi-level governance. J Eur Public Policy 1997;4(4):520–38. https://doi.org/10.1080/ 135017697344046.
- [38] Benz A, Eberlein B. The europeanization of regional policies: patterns of multilevel governance. J Eur Public Policy 1999;6(2):329–48. https://doi.org/ 10.1080/135017699343748.
- [39] Marks G, Hooghe L, Blank K. European integration from the 1980s: state-centric v. multi-level governance. J Common Mark Stud 1996;34(3):341–78. https://doi. org/10.1111/j.1468-5965.1996.tb00577.x.
- [40] Stephenson P. Twenty years of multi-level governance: 'Where Does It Come From? What Is It? Where Is It Going? J Eur Public Policy 2013;20(6):817–37. https://doi.org/10.1080/13501763.2013.781818.
- [41] Follesdal A. Theories of democracy for Europe: multi-level challenges for multi-level governance. SSRN Electron J 2012:1–12. https://doi.org/10.2139/ ssm.1750979.
- [42] Beckmann V, Padmanabhan M. Institutions and sustainability: political economy of agriculture and the environment-essays in honour of konrad hagedorn. Institutions Sustain Polit Econ Agric Environ Honour Konrad Hagedorn 2009: 1–387. https://doi.org/10.1007/978-1-4020-9690-7.
- [43] Jänicke M. The multi-level system of global climate governance the model and its current state. Environ Policy Gov 2017;27(2):108–21. https://doi.org/ 10.1002/eet.1747.

- [44] Brisbois MC. Decentralised energy, decentralised accountability? Lessons on how to govern decentralised electricity transitions from multi-level natural resource governance. Glob Trans 2020;2:16–25. https://doi.org/10.1016/j. elt.2020.01.001.
- [45] Dobravec V, Matak N, Sakulin C, Krajačić G. Multilevel governance energy planning and policy: a view on local energy initiatives. Energy Sustain Soc 2021; 11(1):1–17. https://doi.org/10.1186/s13705-020-00277-y.
- [46] Britton J, Webb J. Institutional work and social skill: the formation of strategic action fields for local energy systems in Britain. Environ Innov Soc Transitions 2024;50:100789. https://doi.org/10.1016/j.eist.2023.100789.
- [47] Lo K, Castán Broto V. Co-benefits, contradictions, and multi-level governance of low-carbon experimentation: leveraging solar energy for sustainable development in China. Glob Environ Chang 2019;59, no. June:101993. https://doi.org/ 10.1016/j.gloenvcha.2019.101993.
- [48] Ringel M. Energy efficiency policy governance in a multi-level administration structure — evidence from Germany. Energ Effic 2017;10(3):753–76. https://doi. org/10.1007/s12053-016-9484-1.
- [49] Smith A. Emerging in between: the multi-level governance of renewable energy in the English regions. Energy Policy 2007;35(12):6266–80. https://doi.org/ 10.1016/j.enpol.2007.07.023.
- [50] Ostrom E. Do institutions for collective action evolve? J Bioecon Apr. 2014;16(1): 3–30. https://doi.org/10.1007/s10818-013-9154-8.
- [51] Ostrom E. Background on the institutional analysis and development framework. Policy Stud J 2011;39(1):7–27. https://doi.org/10.1111/j.1541-0072.2010.00394.x.
- [52] Gagliardi F. Institutions and economic change : A critical survey of the new institutional approaches and empirical evidence 2008;37:416–43. https://doi. org/10.1016/j.socec.2007.03.001.
- [53] McGinnis MD. An introduction to IAD and the language of the Ostrom workshop: a simple guide to a complex framework. Policy Stud J 2011. https://doi.org/ 10.1111/j.1541-0072.2010.00401.x.
- [54] Nikolic I, Ghorbani A. "A method for developing agent-based models of sociotechnical systems A Method for Developing Agent-based Models of Sociotechnical," no. May 2014. 2011. https://doi.org/10.1109/ICNSC.2011.5874914.
- [55] Verhoog R, Ghorbani A, Dijkema GPJ. Modelling socio-ecological systems with MAIA: a biogas infrastructure simulation. Environ Model Softw 2016;81:72–85. https://doi.org/10.1016/j.envsoft.2016.03.011.
- [56] Bauwens T. Explaining the diversity of motivations behind community renewable energy. Energy Policy 2016;93:278–90. https://doi.org/10.1016/j. enpol.2016.03.017.
- [57] Dóci G, Vasileiadou E. 'Let's do it ourselves' individual motivations for investing in renewables at community level. Renew Sust Energ Rev 2015;49:41–50. https://doi.org/10.1016/i.rser.2015.04.051.
- [58] Murphy RO, Ackermann KA. Social value orientation : Theoretical and measurement issues in the study of social preferences. 2014. https://doi.org/ 10.1177/1088868313501745.
- [59] Murphy RO, Ackermann KA, Handgraaf MJJ. Measuring Social Value Orientation 2011;6(8):771–81.
- [60] Kastner I, Matthies E. Energy Research & Social Science Investments in renewable energies by German households : A matter of economics, social influences and ecological concern ? Chem Phys Lett 2016;17:1–9. https://doi.org/10.1016/j. erss.2016.03.006.
- [61] Sütterlin B, Brunner TA, Siegrist M. "Impact of social value orientation on energy conservation in different behavioral domains," pp. 1725–1735. 2013. https://doi. org/10.1111/jasp.12128.
- [62] McGinnis MD, Ostrom E. Social-ecological system framework: initial changes and continuing challenges. Ecol Soc 2014;19(2). https://doi.org/10.5751/ES-06387-190230.
- [63] Geels FW. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res Policy 2002;31(8–9):1257–74. https://doi.org/10.1016/S0048-7333(02)00062-8.
- [64] Ghorbani A, Bravo G. Managing the commons: a simple model of the emergence of institutions through collective action. Int J Commons 2016;10(1):200–19. https://doi.org/10.18352/ijc.606.
- [65] 2011 Bruce. 済無no title No title. J Chem Inf Model 2013;53(9):1689–99. https:// doi.org/10.1017/CBO9781107415324.004.
- [66] D. F. F. F. Intentional agents and goal formation. 1998.
- [67] C. R. R. R. Computational social and behavioral science. New Front Study Soc Phenom Cogn Complexity, Adapt 2016:1–7.
- [68] Railsback SF, Grimm V. Agent-based and individual-based modeling: A practical introduction. Princeton: Princeton University Press; 2012.
- [69] Wilensky U, Rand W. An introduction to agent-based modeling. The MIT Press 2015. https://doi.org/10.2307/j.ctt17kk851.
- [70] K.H. Dam, I. Nikolic, and Z. Lukszo, Agent-based Modelling of socio-technical systems. 2013.
- [71] DeAngelis DL, Grimm V. Individual-based models in ecology after four decades. F1000Prime Rep 2014;6(June). https://doi.org/10.12703/P6-39.
- [72] Bonabeau E. Agent-based modeling: methods and techniques for simulating human systems. Proc Natl Acad Sci USA 2002;99(Suppl. 3):7280–7. https://doi. org/10.1073/pnas.082080899.
- [73] De Wildt TE, Chappin EJL, Van De Kaa G, Herder PM, Van De Poel IR. Energy Research & Social Science Conflicted by decarbonisation : Five types of conflict at the nexus of capabilities and decentralised energy systems identified with an agent-based model. Energy Res Soc Sci 2020;64, no. January:101451. https://doi. org/10.1016/j.erss.2020.101451.

- [74] Yue T, Long R, Chen H, Liu J, Liu H, Gu Y. Energy-saving behavior of urban residents in China: a multi-agent simulation. J Clean Prod 2020;252. https://doi. org/10.1016/j.jclepro.2019.119623.
- [75] De Wildt TE, Boijmans AR, Chappin EJL, Herder PM. An ex ante assessment of value conflicts and social acceptance of sustainable heating systems an agentbased modelling approach. Energy Policy 2021;153, no. March:112265. https:// doi.org/10.1016/j.enpol.2021.112265.
- [76] Busch J, Roelich K, Bale CSE, Knoeri C. "Scaling up local energy infrastructure; An agent-based model of the emergence of district heating networks," *Energy Policy*, vol. 100, no. October 2016, pp. 170–180. 2017. https://doi.org/10.1016/j. enpol.2016.10.011.
- [77] Nava C, Korevaar G, Hansen HH. Agent-based modeling of a thermal energy Transition in the built environment. 2019. https://doi.org/10.3390/en12050856.
- [78] Devia W, Agbossou K, Cardenas A. An evolutionary approach to modeling and control of space heating and thermal storage systems. Energ Buildings 2021;234: 110674. https://doi.org/10.1016/j.enbuild.2020.110674.
- [79] Norouziasl S, Jafari A, Wang C. An agent-based simulation of occupancy schedule in office buildings. Build Environ 2020;186, no. September:107352. https://doi. org/10.1016/j.buildenv.2020.107352.
- [80] Dziedzic JW, Yan D, Sun H, Novakovic V. Building occupant transient agentbased model – movement module. Appl Energy 2020;261(7491):114417. https:// doi.org/10.1016/j.apenergy.2019.114417.
- [81] Mittal A, Krejci CC, Dorneich MC, Fickes D. "An agent-based approach to modeling zero energy communities," *Sol. Energy*, vol. 191, no. December 2018, pp. 193–204. 2019. https://doi.org/10.1016/j.solener.2019.08.040.
- [82] Mohammed NA, Al-Bazi A. Management of renewable energy production and distribution planning using agent-based modelling. Renew Energy 2021;164: 509–20. https://doi.org/10.1016/j.renene.2020.08.159.
- [83] Mittal A, Krejci CC, Dorneich MC, Fickes D. An agent-based approach to modeling zero energy communities. Sol Energy 2019;191:193–204. https://doi.org/ 10.1016/j.solener.2019.08.040.
- [84] Rai V, Robinson SA. Agent-based modeling of energy technology adoption: empirical integration of social, behavioral, economic, and environmental factors. Environ Model Softw 2015;70:163–77. https://doi.org/10.1016/j. envsoft.2015.04.014.
- [85] Burg V, Troitzsch KG, Akyol D, Baier U, Hellweg S, Thees O. Farmer's willingness to adopt private and collective biogas facilities: an agent-based modeling approach. Resour Conserv Recycl 2020;167(December):2021. https://doi.org/ 10.1016/j.resconrec.2021.105400.
- [86] Fraunholz C, Kraft E, Keles D, Fichtner W. Advanced price forecasting in agentbased electricity market simulation. Appl Energy 2020;290(November):116688. 2021, https://doi.org/10.1016/j.apenergy.2021.116688.
- [87] Mahmood I, et al. Modeling, simulation and forecasting of wind power plants using agent-based approach. J Clean Prod 2020;276:124172. https://doi.org/ 10.1016/j.jclepro.2020.124172.
- [88] Fouladvand J. Energy Research & Social Science Why and how can agent-based modelling be applied to community energy systems? A systematic and critical review. *Energy Res* Soc Sci 2024;114, no. April:103572. https://doi.org/10.1016/ j.erss.2024.103572.
- [89] Conradie PD, De Ruyck O, Saldien J, Ponnet K. Who wants to join a renewable energy community in Flanders? Applying an extended model of theory of planned behaviour to understand intent to participate. Energy Policy 2021;151, no. February:112121. https://doi.org/10.1016/j.enpol.2020.112121.
- [90] Tan Y, Ying X, Gao W, Wang S, Liu Z. Applying an extended theory of planned behavior to predict willingness to pay for green and low-carbon energy transition. J Clean Prod 2022;387(March):135893. 2023, https://doi.org/10.1016/j.jclepro. 2023.135893.
- [91] Chen MF. Extending the theory of planned behavior model to explain people's energy savings and carbon reduction behavioral intentions to mitigate climate change in Taiwan-moral obligation matters. J Clean Prod 2016;112:1746–53. https://doi.org/10.1016/j.jclepro.2015.07.043.
- [92] Transition C. SDE ++ 2023 stimulation of sustainable energy production and climate Transition. 2023.
- [93] Magnusson D. Who brings the heat? from municipal to diversified ownership in the Swedish district heating market post-liberalization. Energy Res Soc Sci 2016; 22:198–209. https://doi.org/10.1016/j.erss.2016.10.004.
- [94] Watts DJ, Strogatz SH. Collective dynamics of small-world networks. Nature 1998;393:440–2.
- [95] Jung M, Hwang J. Structural dynamics of innovation networks funded by the European Union in the context of systemic innovation of the renewable energy sector. Energy Policy 2016;96:471–90. https://doi.org/10.1016/j. enpol.2016.06.017.
- [96] Ghorbani A, Nascimento L, Filatova T. Growing community energy initiatives from the bottom up: Simulating the role of behavioural attitudes and leadership in the Netherlands. *Energy Res* Soc Sci 2020;70, no. March:101782. https://doi. org/10.1016/j.erss.2020.101782.
- [97] Societies A, Simulation S. "Which Sensitivity Analysis Method Should I Use for My Agent-Based Model ?," pp. 1–35. 2016. https://doi.org/10.18564/jasss.2857.
- [98] Bouw K, Wiekens CJ, Tigchelaar C, Faaij A. Involving citizens in heat planning: a participatory process Design for Informed Decision-Making. Sustain 2023;15(3). https://doi.org/10.3390/su15031937.
- [99] Bhola P, Chronis A, Kotsampopoulos P. "Business Model Selection for Community Energy Storage : A," pp. 1–30. 2023.
- [100] Kubli M, Puranik S. "A typology of business models for energy communities : Current and emerging design options," *Renew. Sustain. Energy Rev.*, vol. 176, no. August 2022, p. 113165. 2023. https://doi.org/10.1016/j.rser.2023.113165.

J. Fouladvand et al.

- [101] Narjabadifam N, Fouladvand J, Gül M. Critical review on community-shared solar—advantages, challenges, and future directions. Energies 2023;16(8). https://doi.org/10.3390/en16083412.
- [102] Ajzen I. The theory of planned behavior. Organ Behav Hum Decis Process 1991;50 (2):179–211. https://doi.org/10.1016/0749-5978(91)90020-T.
- [103] Westaby JD. "Behavioral reasoning theory : Identifying new linkages underlying intentions and behavior," vol. 98, pp. 97–120. 2005. https://doi.org/10.1016/j. obhdp.2005.07.003.
- [104] Williamson OE. Transaction cost economics: how it works; where it is headed. Economist 1998;146(1):23–58. https://doi.org/10.1023/A:1003263908567.
- [105] D. A. A. A. "Multicriteria approach for a multisource district heating," Green Energy Technol., no. 9783319757735, pp. 21–33. 2018.
- [106] Tsoutsos T, Drandaki M, Frantzeskaki N, Iosifidis E, Kiosses I. Sustainable energy planning by using multi-criteria analysis application in the island of Crete37; 2009, p. 1587–600, https://doi.org/10.1016/j.enpol.2008.12.011.
- 2009. p. 1587–600. https://doi.org/10.1016/j.enpol.2008.12.011.
 [107] Sadiq R, Karunathilake H, Hewage K. Renewable energy selection for net-zero energy communities : Life cycle based decision making under uncertainty 2019; 130:558–73. https://doi.org/10.1016/j.renene.2018.06.086.
- [108] Mckenna R, Bertsch V, Mainzer K, Fichtner W. Combining local preferences with multi-criteria decision analysis and linear optimization to develop feasible energy concepts in small communities. Eur J Oper Res 2018;268(3):1092–110. https:// doi.org/10.1016/j.ejor.2018.01.036.